Author's response to reviews

Title: Buckle Up Safely: A cluster randomised trial to evaluate the effectiveness of a pre-school based program to increase appropriate use of child restraints.

Authors:

Rebecca Q Ivers (rivers@george.org.au)
Lisa Keay (lkeay@georgeinstitute.org.au)
Julie Brown (j.brown@neura.edu.au)
Lynne E Bilston (L.Bilston@neura.edu.au)
Kate Hunter (khunter@georgeinstitute.org.au)
Judy M Simpson (judy.simpson@sydney.edu.au)
Mark Stevenson (Mark.Stevenson@monash.edu)

Version: 2 Date: 4 January 2011

Author's response to reviews:

Revisions as requested by the editor are the addition of a conflict of interest section, and details of human ethics approvals have also been added. There is also a minor change to one reference.